Author’s response to reviews

Title: Tubulointerstitial nephritis complicating IVIG therapy for X-linked agammaglobulinemia

Authors:

Keisuke Sugimoto (ksugimo@med.kindai.ac.jp)
Hitomi Nishi (hitomi1017jiro@yahoo.co.jp)
Tomoki Miyazawa (tmk_3830@yahoo.co.jp)
Norihisa Wada (nmcs@med.kindai.ac.jp)
Akane Izu (akane-momo.0131@hotmail.co.jp)
Takuji Enya (pedi@med.kindai.ac.jp)
Mitsuru Okada (mitsuru@med.kindai.ac.jp)
Tsukasa Takemura (tsukasa@med.kindai.ac.jp)

Version: 6 Date: 17 June 2014

Author’s response to reviews: see over
Dear Prof. Neild,

Thank you for your letter of May 29, 2014 regarding our manuscript (MS: 1365747672121644R2). We appreciate the editor’s and reviewer’s assessment of the work. Please find the revised manuscript enclosed. Corrections were indicated by underline.

Author's response to editor

Tubulointerstitial nephritis complicating IVIG therapy for X-linked agammaglobulinemia (MS: 1365747672121644R2)

Authors:
Sugimoto K: ksugimo@med.kindai.ac.jp,
Nishi H: hitomi1017jiro@yahoo.co.jp,
Miyazawa T: tmk_3830@yahoo.co.jp,
Wada, N: nmcs@med.kindai.ac.jp,
Izu A: akane-momo.0131@hotmail.co.jp,
Enya T: pedi@med.kindai.ac.jp,
Okada M: mitsuru@med.kindai.ac.jp
Takemura T: tsukasa@med.kindai.ac.jp

Dear editor,
Thank you for your suggestion. We corrected numerous grammatical errors and typographical errors. The revised our manuscript was checked by native speaker. All corrections were indicated by underline.
I am very sorry for having trouble with you.
Sincerely,
Tsukasa Takemura